Author's response to reviews

Title: Analysis of the effects of exposure to acute hypoxia on oxidative lesions and tumour progression in a transgenic mouse breast cancer model

Authors:

Tuula M Kalliomäki (tuula.kalliomaki@gmail.com)
Gordon McCallum (gordon.mccallum@utoronto.ca)
Sarah Jane Lunt (sarahjanerutherford@gmail.com)
Peter G Wells (pg.wells@utoronto.ca)
Richard P Hill (hill@uhnres.utoronto.ca)

Version: 7 Date: 9 May 2008

Author's response to reviews:

Font size in Figure 2 increased to show the number of animals used per group (missing for all but two time points despite identical font size in the original submission).